Author's response to reviews

Title: Gene expression profile of cervical and skin tissues from human papillomavirus type 16 E6 transgenic mice

Authors:

D Mendoza-Villanueva (dmendoza@cinvestav.mx)
J Diaz-Chavez (josediaz@cinvestav.mx)
L Uribe-Figueroa (luribe@inmegen.gob.mx)
C Rangel-Escareño (crangel@inmegen.gob.mx)
A Hidalgo-Miranda (ahidalgo@inmegen.gob.mx)
S March-Mifsut (smarch@inmegen.gob.mx)
G Jimenez-Sanchez (dirgeneral@inmegen.gob.mx)
PF Lambert (lambert@oncology.wisc.edu)
P Gariglio (vidal@cinvestav.mx)

Version: 4 Date: 30 October 2008

Author's response to reviews: see over
October 30th, 2008

Editor in Chief
Melissa Norton, MD
BMC Cancer

Dear Melissa Norton:

Thank you for your letter dated 10/17/08, concerning our manuscript: “GENE EXPRESSION PROFILE OF CERVICAL AND SKIN TISSUES FROM HUMAN PAPILLOMAVIRUS TYPE 16 E6 TRANSGENIC MICE” (MS: 1178109161208332) by Mendoza-Villanueva et al. Considering the content of your letter and the suggestions of Editorial production Team, we now submit a revised manuscript version in which we have made the last suggested changes. We include a covering letter with specific arguments for each comment. The changes made in our new manuscript version are described specifically in each answer to the Editorial production team.

We really appreciate the suggestions since they greatly improved our manuscript. We hope that now you would find our paper suitable for publication in your important Journal.

We thank you in advance,

Sincerely yours,

Dr. Patricio Gariglio
Full Professor
Department of Genetics and Molecular Biology
CINVESTAV
Mexico City
Figure order - The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures.

Please list all the figures, for example: Figures 5, 6, 7, 8 and 9.
Currently fig 7 doesn't appear to be referred to in the text. Please amend this.

Answer: We now mentioned the figures in correct order, and all figures submitted are cited including Fig 7.